Author's response to reviews

Title: Centrosome clustering and cyclin D1 gene amplification in double minutes are common events in chromosomal unstable bladder tumors

Authors:

Javier del Rey (javier.delrey@uab.cat)
Esther Prat (eprat@idibell.org)
Immaculada Ponsa (imma.ponsa@uab.cat)
Josep Lloreta (jlloreta@imas.imim.es)
Antoni Gelabert (agelabert@imas.imim.es)
Ferran Algaba (falgaba@fundacio-puigvert.es)
Jordi Camps (campsj@mail.nih.gov)
Rosa Miro (rosa.miro@uab.cat)

Version: 4 Date: 21 May 2010

Author's response to reviews: see over
Dear editor,

Please find enclosed the revised manuscript “Centrosome clustering and Cyclin D1 gene amplification in double minutes are common events in chromosomal unstable bladder tumors” by del Rey et al. All formatting changes requested have been done:

Major revisions:

Authors’ contributions: Author’s contributions section has been formatted according to the BMC Cancer journal’s style.

Table labelling: Heading for Table 1 has been added. Vertical alignment of the text has been changed to horizontal alignment.

Figures: Figure titles have been provided.

Minor revisions:

Title: Title format has been changed

Email address: Email addresses for all the authors has been included on the title page.

Key words: Have been removed.

Abbreviations: Abbreviations have been formatted in to a paragraph.

Typography has been checked.

Sincerely yours,

Rosa Miró, PhD